Author's response to reviews

Title: Leishmania infantum leishmaniasis in corticosteroid - treated patients.

Authors:

   Silvia Pittalis (pittalis@inmi.it)
   Emanuele Nicastri (nicastri@inmi.it)
   Francesco Spinazzola (sspinazz@tin.it)
   Piero Ghirga (p.ghirga@tiscali.it)
   Michele DE Marco (demarco@inmi.it)
   Mariagrazia Paglia (pagliamicromol@inmi.it)
   Pasquale Narciso (narciso@inmi.it)

Version: 5 Date: 16 August 2006

Author's response to reviews:

Dear editor of BMC Infectious Diseases,
Please find enclosed the manuscript entitled "Leishmania infantum leishmaniasis in corticosteroid - treated patients" Manuscript number 1432989538960661 revised according to the last suggestions of the editor.
Thanks in advance for the consideration of our work

Rome, August 18, 2006

Emanuele Nicastri